 Vc/29-Nov-04
Week49 starting 29 Nov 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 29 Nov 04  11/29/2004 3:39:23 PM   
	LHC SM18 Magnet Test Facility 


	4
	A1
	MBBR3017
	PT 12 Warm Up
	212.4

	11
	A2
	MBBL3238
	Prep 0 Scan Bench
	294.6

	12
	B1
	_
	_
	294.7

	9
	B2
	MBAL2120
	Prep 3 CDW1 HV
	296.3

	3
	C1
	MBBR3239
	Prep 5 Pumping + Cool Down
	1.91

	10
	C2
	MBAL3231
	Finished Ready to go Final scan
	296.9

	1
	D1
	SSS030
	Stretched Wire Cold SC25
	1.9

	8
	D2
	SSS040
	ICS 4 Final connection
	295.5

	2
	E1
	MBAL2122
	PT 6.1 Training 1
	1.89

	5
	E2
	MBBL1170
	Prep 5 Pumping + Cool Down
	103.6

	6
	F1
	MBBL2118
	PT 12 Warm Up
	62.4

	7
	F2
	MBBR3240
	ICS 4 Final connection
	294.5


Temperature data are taken from TBSM18 site
   
Coordinator this week: Pierre; MTM support on call :   16-3602 .

12 named  Magnets Tested or 13 Magnet  tests  cycles ( leak on ss614  counted as useful test of cryo integrity):
 10 dipoles( 2 with MM)+ 2 SSS( 0 with MM ) +  1  SSSS     Disconnected after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· SSS Problems ( related to SUN ?? )
1) Sometimes the test files are not generated.

2) Sometimes the file is lost while copying.

3) Sometimes during X-Y measurement, X-results are OK But Y-results are exactly "ZERO" OR vise-versa.

4) Some timeswhile copying the result file, PC gives message as "not sufficient space"
· SHUTDOWN for Water : \
· Cluster A and Current leads /water leak repair 

· BENCH B1 ( 1 week for Q9 mods gone);  Q9  SSSS  tests start?

· AOB
	

	Magnets under Test from 21-11-2004 to 27-11-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL3219 
	[image: image1.png]



	20-11-2004 
	33 h.
	26 h.
	43 h.
	16 h.
	9 h.
	25-11-2004 
	120 hours
	
	

	2
	TBB1
	SSS614 
	[image: image2.png]



	18-11-2004 
	34 h.
	-
	-
	-
	-
	22-11-2004 
	96 hours
	
	

	3
	TBB2
	MBBL2043 
	[image: image3.png]



	16-11-2004 
	37 h.
	38 h.
	96 h.
	19 h.
	5 h.
	24-11-2004 
	192 hours
	
	

	4
	TBC1
	MBAR3247 
	[image: image4.png]



	23-11-2004 
	16 h.
	21 h.
	17 h.
	22 h.
	10 h.
	27-11-2004 
	96 hours
	
	

	5
	TBC1
	MBAR3257 
	[image: image5.png]



	18-11-2004 
	24 h.
	48 h.
	18 h.
	17 h.
	7 h.
	23-11-2004 
	120 hours
	
	

	6
	TBC2
	MBBL1168 
	[image: image6.png]



	19-11-2004 
	24 h.
	46 h.
	25 h.
	15 h.
	9 h.
	24-11-2004 
	120 hours
	
	

	7
	TBD1
	SSS039 
	[image: image7.png]



	12-11-2004 
	73 h.
	25 h.
	101 h.
	11 h.
	26 h.
	22-11-2004 
	240 hours
	
	

	8
	TBD2
	SSS101 
	[image: image8.png]



	17-11-2004 
	55 h.
	47 h.
	51 h.
	9 h.
	38 h.
	26-11-2004 
	216 hours
	
	

	9
	TBE1
	MBBR3259 
	[image: image9.png]



	20-11-2004 
	34 h.
	27 h.
	47 h.
	30 h.
	7 h.
	26-11-2004 
	144 hours
	
	

	10
	TBE2
	MBBL1167 
	[image: image10.png]



	22-11-2004 
	26 h.
	47 h.
	13 h.
	29 h.
	7 h.
	27-11-2004 
	120 hours
	
	

	11
	TBE2
	MBBL3237 
	[image: image11.png]



	16-11-2004 
	22 h.
	50 h.
	14 h.
	15 h.
	29 h.
	22-11-2004 
	144 hours
	
	

	12
	TBF1
	MBBL3216 
	[image: image12.png]



	20-11-2004 
	33 h.
	44 h.
	28 h.
	21 h.
	7 h.
	26-11-2004 
	144 hours
	
	

	13
	TBF2
	MBAL3225 
	[image: image13.png]



	19-11-2004 
	21 h.
	22 h.
	36 h.
	17 h.
	5 h.
	23-11-2004 
	96 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL3219
	12661 A.
(D1-L-CFB)
	12850 A.
[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS614
	[image: image15.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL2043
	12049.7 A.
(D2_L_S01)
	12529 A.
(D1-U-S13)
	12787 A.
(D1-U)
	[image: image16.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAR3247
	11625 A.
(D2_U)
	12144 A.
(D1_U)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAR3257
	12419 A.
(D2 - L)
	12850 A.
[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL1168
	10815 A.
(D2-L)
	12154 A.
(D2-U)
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS039
	11768 A.
(QE-34)
	11750 A.
(QE-34)
	12755 A.
(QE-34)
	12629 A.
(QI_12)
	12848 A.
(QI-34-S3)
	12851 A.
(No Data)
	12851 A.
[image: image20.jpg]



	-
	-
	-
	-
	-
	-

	8
	SSS101
	11416 A.
(QE_34)
	12228 A.
(QI-12)
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR3259
	11710 A.
(D2_U)
	11769 A.
(D1_U)
	12159 A.
(D1-L)
	12298 A.
(D1-L)
	12403 A.
(D1-L)
	12446 A.
(D1_L)
	12593 A.
(D1_L)
	12729 A.
(D1-L)
	12810 A.
(D2_U)
	[image: image22.png]



	-
	-
	-

	10
	MBBL1167
	12504.7 A.
(D1-L)
	12771.6 A.
(D2-L)
	[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBL3237
	12225 A.
(D2-U)
	12659 A.
(D2-L)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL3216
	8807 A.
(D1-U)
	10518 A.
(D2_L)
	12189 A.
(D2_L)
	12279 A.
(D2_U)
	12484 A.
(D2_U)
	12783 A.
(D2_U)
	12848 A.
[image: image25.jpg]



	-
	-
	-
	-
	-
	-

	13
	MBAL3225
	9111 A.
(D2-L)
	11183 A.
(D1-U)
	12239 A.
(D2_U)
	12839 A.
(D2 - U)
	12778 A.
(D2-L)
	12796 A.
(D2-L)
	12850 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-


	From 21-11-2004 to 27-11-2004 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBAL3225 
	False 
	19-11-2004 
	21 h.
	22 h.
	36 h.
	17 h.
	5 h.
	23-11-2004 
	4 days
	6 

	2
	TBF1
	MBBL3216 
	False 
	20-11-2004 
	33 h.
	44 h.
	28 h.
	21 h.
	7 h.
	26-11-2004 
	6 days
	6 

	3
	TBE2
	MBBL1167 
	False 
	22-11-2004 
	26 h.
	47 h.
	13 h.
	29 h.
	7 h.
	27-11-2004 
	5 days
	2 

	4
	TBE2
	MBBL3237 
	False 
	16-11-2004 
	22 h.
	50 h.
	14 h.
	15 h.
	29 h.
	22-11-2004 
	6 days
	2 

	5
	TBE1
	MBBR3259 
	False 
	20-11-2004 
	34 h.
	27 h.
	47 h.
	30 h.
	7 h.
	26-11-2004 
	6 days
	9 

	6
	TBD2
	SSS101 
	False 
	17-11-2004 
	55 h.
	47 h.
	51 h.
	9 h.
	38 h.
	26-11-2004 
	9 days
	2 

	7
	TBD1
	SSS039 
	False 
	12-11-2004 
	73 h.
	25 h.
	101 h.
	11 h.
	26 h.
	22-11-2004 
	10 days
	6 

	8
	TBC2
	MBBL1168 
	False 
	19-11-2004 
	24 h.
	46 h.
	25 h.
	15 h.
	9 h.
	24-11-2004 
	5 days
	2 

	9
	TBC1
	MBAR3247 
	False 
	23-11-2004 
	16 h.
	21 h.
	17 h.
	22 h.
	10 h.
	27-11-2004 
	4 days
	2 

	10
	TBC1
	MBAR3257 
	False 
	18-11-2004 
	24 h.
	48 h.
	18 h.
	17 h.
	7 h.
	23-11-2004 
	5 days
	1 

	11
	TBB2
	MBBL2043 
	True 
	16-11-2004 
	37 h.
	38 h.
	96 h.
	19 h.
	5 h.
	24-11-2004 
	8 days
	3 

	12
	TBB1
	SSS614 
	False 
	18-11-2004 
	34 h.
	230 h.
	-170 h.
	0 h.
	1 h.
	22-11-2004 
	4 days
	0 

	13
	TBA1
	MBBL3219 
	True 
	20-11-2004 
	33 h.
	26 h.
	43 h.
	16 h.
	9 h.
	25-11-2004 
	5 days
	1 


 

from 21-11-2004 to 27-11-2004

2043 had many SPECIAL tests; SS039 on D1 with Shafts; 33219 was 3rd time , now with QAntenna
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _23_A


	TBD2
	 Problem with cryo OK: PLC communication problem


	ACR/MTM
	
	

	Wednesday _24_A


	TBD1
	Launch of HV test at warm failed (Before CT)
	AB/CO
	HV
	HV application re-started then OK

	Wednesday _24_N

Friday _26_A


	TBA2
	Water leak coming from power cable.
	MTM
	
	One leak repaired.

Another one appeared. Test finished like this: intervention planned next Tuesday.

	Wednesday _24_N


	TBA2
	Polarity switch not coming in right position
	MTM
	
	Federico called. He adjusted the position of the switch.

	Friday_26_N


	TBC1
	Too low resistance value of the MCO corrector in the 3247 magnet.: 2.94 Ohms instead of 4.8 Ohms.
	MTM
	
	NC opened. Magnet put in stand by

	Friday_26_N


	TBC2
	Incorrect name of the magnet in the e-traveller: 3132 instead of 3231.
	ICS
	
	Name corrected.

	Friday_26_N


	All 
	Stop of pumping: problem with the BP.
	ACR
	
	Stop during 6 h30

	Saturday 27_M


	TBC2 
	HV test stopped during execution to permit the removal of the 3247 magnet
	
	
	Test re-started after.

	Saturday 27_A


	TBB2 
	After final HV magnet warmed up above 10 K.

But no quench because no powering. 
	ACR
	
	Because pumping was stopped and the magnet was warming up.

	Saturday 27_A


	TBD2
	Wrong value of temperature on SSS40.
	MTM
	
	Federico came and changed the setting (calibration).

	Saturday 27_N


	TBD1 
	De-excitation test on SSS : cannot impose the value of 350 A for the ramp rate. 
	MTM
	Power
	

	Sunday 28_M


	TBF2 
	Leica used for a special test impeded to remove the 1169 magnet
	MTM
	
	TC informed. He asked to wait for the TS/SU team to see if a measurements is running.

	Sunday 28_M


	TBD1 
	MQ strength  with SSW repeated fourth times.

· wire broken,

· SPA

· File not generated 

Results of files lost while copying .

Roll angle measurements not OK.

Incoherent result from X-Y measurements. 
	MTM


	
	The fourth times OK.

Measurement repeated.

	Sunday 28_A


	TBD1 

TBF1
	SPA at 5 kA. 

SPA at 3 kA (2 times)
	
	
	

	Sunday 28_A


	TBA1
	Magnet 3017 measured with Long shaft. Cable connection has to be changed from quench antenna to long shaft.
	MTM
	
	Federico called. He changed the cable connection.


 









